Author's response to reviews

Title: Paediatric UK Demyelinating Disease Longitudinal Study (PUDDLS)

Authors:

Michael Absoud (michaelabsoud@childdemyelination.org.uk)
Carole Cummins (c.l.cummins.20@bham.ac.uk)
Wui K Chong (chongk@gosh.nhs.uk)
Katharine Foster (Katharine.foster@bch.nhs.uk)
Christian Degoede (christian.degoede@lthtr.nhs.uk)
Roxana Gunny (gunnyr@gosh.nhs.uk)
Cheryl Hemingway (heminc@gosh.nhs.uk)
Philip Jardine (philip.jardine@bristol.ac.uk)
Rachel Kneen (Rachel.kneen@alderhey.nhs.uk)
Marcus Likeman (Marcus.likeman@nbt.nhs.uk)
Ming J Lim (ming.lim@gstt.nhs.uk)
Naomi Sibtain (Naomi.sibtain@nhs.net)
Mike Pike (mike.pike@orh.nhs.uk)
William P Whitehouse (William.whitehouse@nottingham.ac.uk)
Evangeline Wassmer (Evangeline.wassmer@bch.nhs.uk)

Version: 2 Date: 6 June 2011

Author's response to reviews:

Dear BMC Pediatrics Editor,

The manuscript has been checked, and minor changes including spelling corrections and improved layout have been applied.